Lipegfilgrastim for prophylaxis of chemotherapy-induced neutropenia in Dutch patients

J.N.H. Timmer-Bonte^{1,a}*, J. Ouwerkerk², L.M. Faber³, L.G.M. Kerkhofs⁴, L. Laterveer⁵, D. ten Oever⁶, B.P. van Rees⁷, P.W. van der Linden⁸

^aAlexander Monro Hospital, Bilthoven, the Netherlands; ^aCurrent affiliation: Department of Medical Oncology, Radboudumc, Nijmegen, the Netherlands; ^aDepartment of Medical Oncology, Leiden University Medical Center, Leiden, the Netherlands; ^aDepartment of Internal Medicine, Red Cross Hospital, Beverwijk, the Netherlands; ^aDepartment of Internal Medicine, IJsselland Hospital, Capelle aan den IJssel, the Netherlands; ^sDepartment of Internal Medicine, Isala Hospital, Zwolle, the Netherlands; ⁶Department of Medical Oncology, Noordwestziekenhuisgroep, Alkmaar, the Netherlands; ⁷Department of Internal Medicine, Tjongerschans Hospital, Heerenveen, the Netherlands; ⁸Department of Medical Oncology, Spaarne Gasthuis, Haarlem, the Netherlands. *Corresponding author: anja.timmer-bonte@radboudumc.nl

ABSTRACT

Background: Chemotherapy (CT)-induced neutropenia and febrile neutropenia (FN) can lead to changes in the treatment plan, potentially worsening the cancer outcome. This study evaluated the effect of the glycopegylated granulocyte-colony stimulating factor lipegfilgrastim, used as primary (PP) or secondary prophylaxis (SP), on treatment modifications in adult patients receiving cytotoxic CT with or without biological/targeted therapy (BT) for solid and haematological tumours.

Methods: This phase 4, prospective, observational study was conducted in eight centres in the Netherlands, in 2015-2017. Other study objectives were to characterise the population of cancer patients receiving lipegfilgrastim, to evaluate the incidence of CT-induced neutropenic events, and to assess safety.

Results: Of 142 patients, 73.94% had breast cancer and 55.63% received CT in the adjuvant setting. Most patients received lipegfilgrastim as PP (74.65%) and were at low (34.51%) or high risk (39.44%) of FN. CT dose delays were recorded for 22.64% and 36.11% of patients receiving lipegfilgrastim for PP and SP, respectively. CT dose reductions were recorded for 2.11% of patients; no CT dose omissions and one BT dose omission occurred. FN and grade III/IV neutropenia were reported for 5.63% and 9.86% of patients, respectively; associated hospitalisations were rare. The most frequently lipegfilgrastim-related adverse events (AE) were myalgia, bone pain,

and back pain. Serious AEs (55) were reported for 30 (21.13%) patients. There were two deaths, unrelated to lipegfilgrastim administration.

Conclusion: Administration of lipegfilgrastim in routine clinical practice in the Netherlands results in limited CT/ BT dose modifications and low incidence of neutropenic events, with no new safety concerns.

KEYWORDS

Chemotherapy dose modification, chemotherapy-induced neutropenia, febrile neutropenia, lipegfilgrastim, real-world evidence

INTRODUCTION

Occurrence of neutropenia is a well-known risk of chemotherapy (CT), and in particular, severe and febrile neutropenia (FN) can significantly impact the outcome of cancer treatment, resulting in an increase in infectionassociated morbidity and mortality.¹ Moreover, neutropenic events may lead to changes in the treatment plan, such as dose reductions and/or delays in subsequent cycles or even switching to less effective treatment options, ultimately decreasing overall response and life expectancy.^{1,2}

To reduce the risk of CT-induced neutropenia, preventive treatment is required and granulocyte-colony stimulating

© MacChain. All rights reserved.

factors (G-CSFs) have been recognised as effective therapies³ as they stimulate neutrophil production and differentiation.⁴ In a meta-analysis of randomised clinical trials, administration of G-CSFs as primary prophylaxis (PP) was shown to also reduce all-cause mortality compared with patients not receiving G-CSFs, with the decrease in risk varying with CT dose and schedule.5 Current European guidelines recommend G-CSFs support for all patients receiving a CT regimen associated with a high (> 20%) or intermediate (10-20%) risk of FN, for dose-dense or dose-intense CT strategies, or in subsequent CT cycles following a FN episode. Recentlyidentified factors such as older age (≥ 65 years) or low neutrophil count should also be included in an assessment of risk carried out before each cycle.6 Real-world data on the routine use of G-CSFs prophylaxis for CT-induced neutropenia and FN are now available from the United States, where an increase in the use of G-CSFs from 2010 to 2016 led to a decrease in the incidence of FN among cancer patients at intermediate and high risk in the first cycle of CT, despite a clear underutilisation of the recommended prophylaxis.7

Lipegfilgrastim (Teva Pharmaceutical Industries Ltd, Israel), a glycopegylated G-CSF, has previously been shown to reduce the duration of neutropenia and incidence of FN in adult cancer patients treated with cytotoxic CT⁸⁻¹¹ and was approved for use in the European Union in July 2013.¹² In a large prospective, non-interventional study conducted in several European countries, the effectiveness and safety of lipegfilgrastim as PP or secondary prophylaxis (SP) in real-world settings was evaluated in adult patients with different tumour types receiving cytotoxic CT.^{13,14} Here we report data on treatment delays and modifications and incidence of neutropenic events in cancer patients in the Netherlands who received lipegfilgrastim in routine clinical settings.

METHODS

Study design and participants

This phase 4, prospective, observational study was conducted in eight centres in the Netherlands between October 2015 and August 2017. Male and female patients who received cytotoxic CT with or without biological/targeted therapy (BT) for solid and haematological malignant cancer in real-world settings were eligible for enrolment in the study. Lipegfilgrastim was administered as prescribed by the treating physician and according to medical indication, independent of the intention to include the patient into the study. Inclusion criteria were age \geq 18 years and administration of lipegfilgrastim as PP or SP (following a neutropenic event in a previous CT cycle) of CT-induced neutropenia. Patients were excluded from

study enrolment if they had chronic myeloid leukaemia and myelodysplastic syndromes, were pregnant or lactating women, or if they received investigational products as part of a clinical trial.

The study was conducted in full accordance with the Helsinki Declaration and the guidelines for Good Clinical Practice, Good Epidemiological Practice, Good Pharmacoepidemiology Practices, and Good Pharmacovigilance Practices and was approved by institutional ethics committees at each site. All patients signed an informed consent form prior to enrolment in the study.

Patients were followed from the first use of lipegfilgrastim until 6-8 weeks after the last dose. Data pertaining to patient demographics and baseline characteristics, FN risk factors, CT/BT data (dose delay, reduction, omission), incidence of FN or neutropenia, FN-related hospitalisation and use of anti-infectives and antimycotics, and safety were recorded on electronic case report forms. In agreement with the European Society for Medical Oncology (ESMO) guidelines, FN was defined as an oral temperature of > 38.3 °C or two consecutive readings of > 38.0 °C for two hours and an absolute neutrophil count of < 0.5 × 10⁹/l, or expected to fall below 0.5 × 10⁹/l.¹⁵

Study objectives

The primary objective of the study was to describe the effect of lipegfilgrastim used in prophylaxis on CT dose modifications, including delay, omissions, and reduction of CT dose administration, in patients receiving cytotoxic CT for solid and haematological cancers, according to routine clinical practice in the Netherlands. Secondary objectives included the description of cancer patients treated prophylactically with lipegfilgrastim in terms of tumour type and stage, CT setting, demographic characteristics, comorbidities and FN risk, and assessment of safety.

Safety assessments

Adverse events (AEs) and serious AEs (SAEs) were coded with the Medical Dictionary for Regulatory Affairs version 20.0 and summarised by preferred term. All AEs were classified by severity as grade I (mild), 2 (moderate), 3 (severe or medically significant), 4 (life-threatening, urgent intervention needed), and 5 (fatal), according to the National Cancer Institute Common Terminology Criteria for Adverse Events, version 4.0.¹⁶ The relationship of AEs to lipegfilgrastim administration (not related/related) was assessed by the investigators.

Statistical analyses

A sample size of 150 participants was estimated based on feasibility considerations in the entire European population.

Analyses were carried out on the full analysis set, which included all enrolled patients meeting eligibility criteria. Patients were stratified by subgroups, as follows: type of prophylaxis (PP versus SP), tumour type, tumour stage, CT regimen use (adjuvant vs metastatic), gender, age, Eastern Cooperative Oncology Group (ECOG) performance status, and FN risk (low, intermediate, and high).

No imputation of missing values was performed. Analyses were mainly descriptive and interpreted in an explorative way.

RESULTS

Table 1. Participant characteristics at enrolment(full analysis set)						
Characteristic	Safety set (N=142)					
Mean age ± SD, years	57.27 ± 12.09					
Female, n (%)	122 (85.92)					
Primary tumour, n (%)						
Breast	105 (73.94)					
Lymphoma	26 (18.31)					
Haematological malignancies	3 (2.11)					
Prostate	3 (2.11)					
Ovary	2 (1.41)					
Pancreas	2 (1.41)					
Stomach	1 (0.7)					
Risk factor for FN						
Any factor present	140 (98.59)					
Female gender	122 (88.03)					
Age > 65 years	42 (29.58)					
History of prior FN	25 (17.61)					
Advanced disease	12 (8.45)					
Poor nutritional status	4 (2.82)					
Poor performance status	3 (2.11)					
Other	3 (2.11)					
No factor present	2 (1.41)					

FN = febrile neutropenia; n (%) = number (percentage) of patients with the specified characteristic; N = number of patients; SD = standard deviation

Patient characteristics

In total, 144 patients were enrolled in the study and 142 were evaluated (two patients did not receive lipegfilgrastim during the study period). All 142 patients received at least

one dose of lipegfilgrastim followed by a post-cycle efficacy evaluation.

The patients' baseline characteristics are presented in table 1. The mean age was 57 years, with 49 (34.51%) patients being aged \geq 65 years (table 1). Most patients (94.37%) were Caucasian and the large majority (85.92%) were women. Lipegfilgrastim was received as PP for 106 (74.65%) patients and as SP for 36 (25.35%) in 596 cycles (there were a total of 613 cycles for the entire study population). Lipegfilgrastim was administered as SP due to the occurrence of FN (for 15 patients) or neutropenia (in 30 patients), with most patients starting to receive lipegfilgrastim as SP in the second (10, 7.04%) or third (15, 10.56%) CT cycle.

Forty-nine (34.51%) patients were classified as low risk, 37 (26.06%) as intermediate risk, and 56 (39.44%) as high risk of FN. The most frequent individual risk factors for FN were female gender (122 patients, 88.03%) and age > 65 years (42 patients, 29.56%) (table I). Two participants (I.4%) had liver disease and 8 (5.6%) had cardiovascular disease.

CT was planned for all patients, with most participants receiving it in the adjuvant (79 patients,55.63%) and neo-adjuvant (51 patients, 35.92%) setting. Eight patients (5.63%) received CT in a metastatic setting, 2 (1.41%) patients as maintenance, and 2 (1.41%) as other setting. For 28 patients (19.72%), BT was also planned. Most patients received docetaxel/doxorubicin/ cyclophosphamide (37 patients, 26.06%), or doxorubicin/ cyclophosphamide (37 patients, 26.06%), or doxorubicin/ cyclophosphamide/vincristine with or without prednisone (17 patients, 11.97%) as CT treatment.

Chemotherapy dose modifications

Overall, 607 administered cycles were analysed, and the delay of CT ranged between 0 and 3.5 days, with a mean delay of 0.8 ± 3.34 days. The mean delay period was higher in patients using lipegfilgrastim as SP than as PP (I.17 versus 0.71 days) and in patients aged \geq 65 years compared with those aged < 65 years (table 2). Overall, delays were recorded for 37 (26.06%) patients, with higher proportion of patients reporting delays in SP versus PP, blood tumour versus breast tumour, and intermediate/low risk versus high risk of FN subgroups (table 2).

Dose reductions were recorded for 3 (2.11%) female patients with breast cancer using lipegfilgrastim as PP (one reduction each in cycle I, 2, and 4). All three had an Eastern Cooperative Oncology Group (ECOG) status < 2 and a low risk of FN.

No CT dose omissions occurred.

No dose reduction of BT treatment was reported in any patient. Only one BT treatment omission was recorded, for a male patient aged ≥ 65 years, receiving lipegfilgrastim

	n (%)	Number of observations	Mean CT delay ± SD (min-max) (days)
Type of prophylaxis			(initritiax) (days)
Primary	24 (22.64)	487	0.71 ± 3.32 (0-35)
Secondary	13 (36.11)	120	1.17 ± 3.04 (0-21)
Tumour type			
Haematological malignancies	2 (66.67)	19	2.32 ± 7.06 (0-30)
Breast	25 (23.81)	440	0.47 ± 2.33 (0-28)
Lymphoma	9 (34.62)	119	1.55 ± 4.64 (0-35)
Pancreas	1 (50.00)	8	6.13 ± 8.72 (0-21)
CT setting			
Adjuvant	24 (30.38)	340	1.04 ± 3.94 (0-35)
Neo-adjuvant	9 (17.65)	216	0.21 ± 1.13 (0-8)
Metastatic	1 (25.00)	30	0.93 ± 3.24 (0-16)
Maintenance	2 (50.00)	12	4.08 ± 7.59 (0-21)
Other	1 (50.00)	9	1.00 ± 2.12 (0-6)
Gender			
Male	7 (35.00)	91	1.89 ± 5.88 (0-35)
Female	30 (24.59)	516	0.60 ± 2.61 (0-28)
Age			
< 65 years	23 (24.73)	405	0.66 ± 3.06 (0-30)
≥ 65 years	14 (28.57)	202	1.08 ± 3.82 (0-35)
ECOG status			
<2	35 (25.36)	593	0.71 ± 3.02 (0-30)
≥2	2 (50.00)	14	4.50 ± 9.35 (0-35)
FN risk			
Low	17 (34.69)	190	0.97 ± 3.63 (0-30)
Intermediate	11 (29.73)	146	1.42 ± 4.47 (0–35)
High	9 (16.07)	271	0.34 ± 2.14 (0–23)

	Table 2.	Summary	of ch	iemotherap	y treatment	delays,	by subgroup) (full and	ilysis set,	n = 14	12
--	----------	---------	-------	------------	-------------	---------	-------------	-------------	-------------	--------	----

CT = chemotherapy; ECOG = Eastern Cooperative Oncology Group; FN = febrile neutropenia; n (%) = number (percentage) of patients with CT treatment delays in each subgroup; N = number of patients included in the analysis; SD = standard deviation

Note: Subgroups with no CT treatment delays were not presented.

as PP; the patient had lymphoma, used CT in the adjuvant setting, and presented intermediate risk of FN.

Incidence of neutropenia and hospitalisation/treatment

Overall, 8 (5.63%) patients reported FN. The majority (six) were female with an ECOG score < 2 and three received lipegfilgrastim as PP. FN was reported in four patients with lymphoma, two with breast cancer, one with pancreatic

cancer, and one with stomach cancer. Most participants with reported FN used adjuvant CT (five patients) and were at low (three patients) or intermediate (four patients) risk of FN. Grade III/IV neutropenia at a given time were recorded in 14 (9.86%) patients, with 9 (6.34%) and 5 (3.52%) patients reporting grade III and grade IV neutropenia, respectively. Four (2.82%) patients were hospitalised due to FN, 2 (1.41%) due to CT-induced neutropenia; 39 (27.46%) patients were hospitalised for other reasons, not related to neutropenic events.

Anti-infective treatment was used in 54 (38.03%) patients, with a mean duration of 13.97 ± 16.84 days. For 5 (3.52%) and 1 (0.70%) of these patients, the use was related to FN and CT-induced neutropenia, respectively (Supplementary material, table SI)*. Five (3.52%) patients were administered intravenous anti-infectives for FN.

Antimycotic treatment was used a total of 49 times in 23 (16.31%) patients (Supplementary material, table S2)* and the mean duration was 27.41 ± 28.72 days; the treatment was not related to FN and CT-induced neutropenia for any of the patients.

Safety

At least one treatment-emergent AE (TEAE) occurred in all patients, the most frequently reported being myalgia (22.53%), bone pain (17.61%), pyrexia (14.79%), and back pain (11.27%). Most TEAEs (88.66%) were mild or moderate in nature and 41.06% were considered related to study medication. Myalgia (19.01%), bone pain (16.20%), back pain (7.75%), and arthralgia (5.63%) were the most frequent TEAEs related to lipegfilgrastim administration. Three patients (2%) discontinued lipegfilgrastim prematurely due to adverse events.

Overall, 55 SAEs were recorded in 30 (21.13%) patients, with the most frequent being pyrexia and pneumonia (each in seven cases), and FN and lung infection (each in four cases); most of them were mild to moderate. In total, 14 SAEs in 10 (7.04%) patients were considered as possibly related to lipegfilgrastim administration.

Fatal events were reported for 2 (I.41%) patients, with the AEs leading to death being advanced disease and cancer progression for one patient and ureteral stent infection (urosepsis) for the other. Neither deaths were considered related to lipegfilgrastim.

DISCUSSION

This is the first study assessing CT dose modifications following administration of lipegfilgrastim as prophylaxis for CT-induced neutropenic events in real-world settings in Dutch patients. Our findings confirm effectiveness and raises no new concerns about safety when compared to clinical trial results.

This study showed that dose delay was the most common modification of CT treatment following lipegfilgrastim administration, occurring in 26.06% of patients, while CT dose reductions were rare, reported for only 2.11% of patients. No CT dose omissions occurred. Dose delays were reported more frequently in patients receiving lipegfilgrastim as SP than those receiving it as PP (36.11% versus 22.64%).

In the real-world prospective study assessing the same objectives in a larger, European population which included Dutch patients, slightly lower rates were observed for overall dose delays, reductions, and omissions, but similar to our study, dose modifications were reported with higher frequency, when lipegfilgrastim was used for SP rather than for PP (for 28.1% and 20.1% of patients).14 In contrast to our observations, dose modifications in Belgian patients occurred in 33.3% and 52.4% of patients receiving lipegfilgrastim as PP and SP, respectively, with dose reductions being more common. Of note, most Belgian patients had breast cancer and more than half (54.7%) received dose-dense regimens, unlike in our study.17 In another non-interventional study conducted in 2489 German patients undergoing CT in routine clinical practice (NADIR), dose delays were observed at lower rates than in the current study for 11.2% of patients.¹⁸ However, in the NADIR study, 16.3% of patients received lipegfilgrastim as SP, compared to 25.35% of the Dutch population, which may constitute one of the possible explanations for the difference in dose delay rates. In contrast, dose reductions were reported more frequently in the NADIR study in 19.6% of patients.¹⁸ However, these comparisons are hindered by differences in the characteristics of patients in each study, which varied, also in terms of tumour type, stage, metastasis, age, performance status, or FN risk. Of note, a higher percentage of Dutch patients were considered at low risk of FN (34.51%), compared with < 10% in the NADIR study.¹⁸ In clinical trial settings, the rates of dose delays ranged from 16.2%¹⁹ to 30.7%⁸ of patients in the first cycle after initial administration of lipegfilgrastim, while no dose omissions/reductions were reported, suggesting that dose modifications can vary greatly with demographics, tumour type, and prescribed CT.

Overall, FN was observed in 5.63% of patients in our study and grade III and IV neutropenia occurred in 6.34% and 3.52% of patients. Compared to our study, a lower incidence (2.7%) was observed in the NADIR study for FN and higher rates of severe neutropenia (26.8%) were reported.¹⁸ Despite differences in the proportion of patients at low risk of FN and lipegfilgrastim prophylactic administration for the two populations, both studies show that lipegfilgrastim is effective in reducing CT-induced neutropenic events and further confirm effectiveness data observed in phase III clinical trials.^{8,9,19} Randomised head-to-head trials have been conducted between lipegfilgrastim and pegfilgrastim. A meta-analysis concludes lipegfilgrastim showed a lower, nonsignificant risk of febrile neutropenia compared with pegfilgrastim (risk ratio (RR)=0.34, 95% CI: 0.05, 2.14). Lipegfilgrastim has a statistically significantly shorter absolute neutrophil count recovery time without significant differences in bone pain.²⁰ Moreover, in the current study, the rates of FN/neutropenia-related hospitalisations, intensive care unit stays, and use of anti-infectives/antimycotics were low, suggesting that the associated cost can be decreased considerably when lipegfilgrastim is used as prophylactic treatment of CT-induced neutropenic events. According to several modelling studies, lipegfilgrastim was estimated to be a likely cost-effective alternative to other G-CSFs in patients with breast cancer.^{21,22}

Lipegfilgrastim was well tolerated, with a safety data comparable to that observed in the NADIR study. A comparable proportion of Dutch and German patients experienced SAEs (21.13% versus 18.0%, respectively), and none of the fatal events occurring during the two studies were related to lipegfilgrastim administration.¹⁸ Myalgia, bone pain, and back pain were the most frequent lipegfilgrastim-related TEAEs in the current study, consistent with observations from real-life studies and the safety profile assessed in pivotal clinical trials.^{12,18} Of note, in the NADIR study, patients evaluated the administration as easy to handle and the most frequently documented reason for discontinuing lipegfilgrastim prematurely was that prophylaxis with lipegfilgrastim was no longer considered necessary.¹⁸

Our study has several potential limitations. Specific reasons for changes in CT treatment were not collected and therefore, we cannot conclude on a potential association between lipegfilgrastim administration and CT treatment modifications. Analyses were mainly descriptive and the sample size was relatively low, although patients from eight centres in the Netherlands were enrolled in the study which is likely to ensure a representative sample. Selection bias and confounding by indication are potential biases,

REFERENCES

- Lalami Y, Klastersky J. Impact of chemotherapy-induced neutropenia (CIN) and febrile neutropenia (FN) on cancer treatment outcomes: An overview about well-established and recently emerging clinical data. Crit Rev Oncol Hematol. 2017;120:163-79.
- 2. Lyman GH. Impact of chemotherapy dose intensity on cancer patient outcomes. J Natl Compr Canc Netw. 2009;7:99-108.
- Mhaskar R, Clark OA, Lyman G, Engel Ayer Botrel T, Morganti Paladini L, Djulbegovic B. Colony-stimulating factors for chemotherapy-induced febrile neutropenia. Cochrane Database Syst Rev. 2014;CD003039.
- Petros W. Colony-stimulating factors. In: Chabner B, Longo D, editors. Cancer chemotherapy and biotherapy - Principles and practice. 3rd ed. Philadelphia, PA, USA: Lippincott Williams & Wilkins; 2001. p. 357-92.
- Lyman GH, Dale DC, Culakova E, et al. The impact of the granulocyte colony-stimulating factor on chemotherapy dose intensity and cancer survival: a systematic review and meta-analysis of randomized controlled trials. Ann Oncol. 2013;24:2475-84.
- Aapro MS, Bohlius J, Cameron DA, et al. 2010 update of EORTC guidelines for the use of granulocyte-colony stimulating factor to reduce the incidence of chemotherapy-induced febrile neutropenia in adult patients with lymphoproliferative disorders and solid tumours. Eur J Cancer. 2011;47:8-32.
- Weycker D, Bensink M, Lonshteyn A, Doroff R, Chandler D. Use of colony-stimulating factor primary prophylaxis and incidence of febrile

but the majority of the included patients were at risk of FN (table 1) and lipegfilgrastim support was according to current guidelines.^{5,6} Data in the entire European population will soon be available and will provide an improved interpretation of the results for the Dutch patients in the larger context of clinical practice in Europe. In conclusion, the use of lipegfilgrastim as either PP or SP during cytotoxic CT treatment led to a relatively low proportion of patients with CT dose modifications or with neutropenic events, in real-world settings in the Netherlands. Administration of lipegfilgrastim was well tolerated, with no new safety concerns arising during the study.

A C K N O W L E D G E M E N T S

The authors are grateful to the patients and all hospital staff involved in the study. The authors also thank Petronela M. Petrar (Modis) for medical writing and publication coordination services.

DISCLOSURES

The authors report no conflicts of interest.

This work was supported by Teva Pharmaceuticals. All costs associated with the development and publication of the manuscript were covered by Teva Nederland BV.

*Supplementary materials are available from the authors upon request.

neutropenia from 2010 to 2016: A longitudinal assessment. Curr Med Res Opin. 2019;35:1037-80.

- Bondarenko I, Gladkov OA, Elsaesser R, Buchner A, Bias P. Efficacy and safety of lipegfilgrastim versus pegfilgrastim: A randomized, multicenter, active-control phase 3 trial in patients with breast cancer receiving doxorubicin/docetaxel chemotherapy. BMC Cancer. 2013;13:386.
- Buchner A, Elsasser R, Bias P. A randomized, double-blind, active control, multicenter, dose-finding study of lipegfilgrastim (XM22) in breast cancer patients receiving myelosuppressive therapy. Breast Cancer Res Treat. 2014;148:107-16.
- Buchner A, Lammerich A, Abdolzade-Bavil A, Muller U, Bias P. Lipegfilgrastim: pharmacodynamics and pharmacokinetics for body-weight-adjusted and 6 mg fixed doses in two randomized studies in healthy volunteers. Curr Med Res Opin. 2014;30:2523-33.
- Volovat C, Bondarenko I, Gladkov O, et al. Efficacy and safety of lipegfilgrastim compared with placebo in patients with non-small cell lung cancer receiving chemotherapy: Post hoc analysis of elderly versus younger patients. Support Care Cancer. 2016;24:4913-20.
- European Medicines Agency. Lonquex. Summary of product characteristics 2013 [Internet, accessed December 12, 2019]. Available from: http://www.ema.europa.eu/documents/product-information/ lonquex-epar-product-information_en.pdf.

- 13. Pichler P, Claes N, Mazza P, et al. Use of lipegfilgrastim in clinical practice for the prophylaxis of chemotherapy-induced neutropenia: Interim results of pan-European non-interventional study. Ann Oncol. 2016;27:1459P.
- 14. Steger G, Pichler P, Airoldi M, et al. Use of lipegfilgrastim for the prophylaxis of chemotherapy-induced neutropenia: Pan-European non-interventional study. Ann Oncol. 2018;29:viii603-viii40.
- Klastersky J, de Naurois J, Rolston K, et al. Management of febrile neutropaenia: ESMO Clinical Practice Guidelines. Ann Oncol. 2016;27:v111-v8.
- National Cancer Institute Common Terminology Criteria for Adverse Events, version 4.0 [Internet, accessed December 12, 2019]. Available from: http://view.officeapps.live.com/op/view.aspx?src=https://ctep. cancer.gov/protocolDevelopment/electronic_applications/docs/Reverse_ Mapping_CTCAE_4_to_CTCAE_3.xls.
- Fontaine C, Claes N, Graas M, Kargar Samani K, Vuylsteke P, Vulstekef C. Effect of lipegfilgrastim administration as prophylaxis of chemotherapyinduced neutropenia on dose modification and incidence of neutropenic events: Real-world evidence from a non-interventional study in Belgium and Luxembourg. Acta Clin Belg. 2019:1-6.

- Fietz T, Luck A, Schulz H, et al. Prophylaxis of chemotherapy-induced neutropenia and febrile neutropenia with lipegfilgrastim in 2489 cancer patients: final results from the non-interventional study NADIR. Curr Med Res Opin. 2019;35:1127-38.
- Gladkov OA, Buchner A, Bias P, Muller U, Elsasser R. Chemotherapyassociated treatment burden in breast cancer patients receiving lipegfilgrastim or pegfilgrastim: Secondary efficacy data from a phase III study. Support Care Cancer. 2016;24:395-400.
- 20. Bond TC, Szabo E, Gabriel S et al. Meta-analysis and indirect treatment comparison of lipegfilgrastim withpegfilgrastim and filgrastim for the reduction of chemotherapy-induced neutropenia-related events. Oncol Pharm Pract. 2018;24(6):412-23.
- 21. Akpo EIH, Jansen IR, Maes E, Simoens S. Cost-utility analysis of lipegfilgrastim compared to pegfilgrastim for the prophylaxis of chemotherapy-induced neutropenia in patients with stage II-IV breast cancer. Front Pharmacol. 2017;8:614.
- 22. Gao L, Li SC. Cost-effectiveness analysis of lipegfilgrastim as primary prophylaxis in women with breast cancer in Australia: A modelled economic evaluation. Breast Cancer. 2018;25:671-80.